Title: Androgen secreting steroid cell tumor of the ovary in a young lactating woman with acute onset of severe hyperandrogenism: a case report and review of literature

Authors:

Altaf Gauhar (altafhaji786@yahoo.com)
Shekhar Sharma (drshekharsharma@gmail.com)
Manoj Babu (manojbabur@aims.amrita.edu)
Vijaykumar D K (dkvijaykumar@aims.amrita.edu)
K Chithrathara (chithrathara@aims.amrita.edu)

Version: 12 Date: 9 December 2007

Author's response to reviews: see over
Dear Editor

Thank you very much for your suggestions and corrections. We have made following changes as suggested by you. The point to point explanation is as follows:

1. Title page now includes the manuscript title (in sentence case), the full names of authors (on one line) and beneath that, associated institutional affiliations for all authors (denoted by superscript numbers). The corresponding author details are then listed, and a list of all author email addresses following that. The corresponding author is denoted with the use of * in the author. The title page is now aligned to the left, and there are no superfluous extras. The short/running titles are also removed.

2. Affiliations – All the authors are affiliated to the same institution and one department. It is now written in full and is as per the suggested format i.e. department, institute, city, country.

3. Each author is linked to his or her corresponding affiliation with the use of superscript numbers.

4. The authors' qualifications/titles are removed from the manuscript and submissions system.

5. Abstract – It is identical in both the manuscript and submission system.

6. Title – It is identical in both the manuscript and submission system.

7 Title – The full stop is removed from the end of the title.

8. Section headings - Sentence case is used for all headings and sub-headings in the manuscript. All headings and titles are aligned to the left and underlining is removed.

9. CASE REPORT is changed to Case presentation and CONSENT SECTION to Consent

10. SOURCES OF FUNDING is removed and is placed into the acknowledgements.

11. The font is made consistent throughout the document.

12. Figure legends section is included after the references and a description of each figure is placed there.

13. Table 1 is placed after the figure legends.

14. The information is removed from the top of the second page.

15. The numbering is removed from the authors' contributions section and is formatted as per the suggested format.

16. AUTHORS CONTRIBUTIONS is changed to Authors' contributions

17. The types of bracket are are replaced from ( ) to [ ] throughout the document.

Thanks,

Dr. Altaf Gauhar Haji